Author’s response to reviews

Title: Significance of the parkin and PINK1 gene in Jordanian families with incidences of young-onset and juvenile parkinsonism

Authors:

Ronny Myhre (ronny.myhre@ntnu.no)
Stina Steinkjer (stina.steinkjer@gmail.com)
Alice Stormyr (alsto@hotmail.com)
Gina L Nilsen (gina.l.n@gmail.com)
Hiba A Zayyad (hiba_zayad@yahoo.com)
Khalid Horany (khalidhorany@hotmail.com)
Mohamad K Nusier (mick@just.edu.jo)
Helge Klungland (helge.klungland@ntnu.no)

Version: 4 Date: 11 November 2008

Author’s response to reviews: see over
Dear Editor,

Please find attached our manuscript, Myhre et al., “Significance of the parkin and PINK1 gene in Jordanian families with incidences of young-onset and juvenile parkinsonism”, including a statement in the method section on informed consent.

Our research project was approved of the research ethics committee for research on human, headed by the dean of Medical School Jordan. An informed consent was obtained from each patient participated in the study in compliance with the Helsinki Declaration.

We state that all blood samples were collected after gathering informed consent.

Corrected from:

“Genetic studies: DNA was collected from all available family members. Blood samples were collected on EDTA tubes and DNA was extracted using Wizard® DNA Extraction kit (Promega).”

Corrected to:

“Genetic studies: DNA was collected from all available family members. Blood samples were collected on EDTA tubes after gathering informed consent from each participant in compliance with the Helsinki Declaration. DNA was extracted using Wizard® DNA Extraction kit (Promega).”

Please contact me for future correspondence.

Sincerely,
Ronny Myhre